#### AGUIDO ADVANCING EARTH AND SPACE SCIENCE

# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2019GL082459

#### **Key Points:**

- Soil respiration increases with proximity to temperature forest edges
- Edge soil respiration enhancements are driven by large positive gradients in soil temperature
- Forest edge carbon dynamics are generally omitted from ecosystem models and carbon accounting frameworks

**Supporting Information:** 

Supporting Information S1

#### Correspondence to:

I. A. Smith, iasmith@bu.edu

#### **Citation**:

Smith, I. A., Hutyra, L. R., Reinmann, A. B., Thompson, J. R., & Allen, D. W. (2019). Evidence for edge enhancements of soil respiration in temperate forests. *Geophysical Research Letters*, 46. https://doi.org/10.1029/ 2019GL082459

Received 13 FEB 2019 Accepted 12 APR 2019 Accepted article online 17 APR 2019

# Dessived 12 FEB 2010

©2019. American Geophysical Union. All Rights Reserved.

# **Evidence for Edge Enhancements of Soil Respiration in Temperate Forests**

Ian A. Smith<sup>1</sup>, Lucy R. Hutyra<sup>1</sup>, Andrew B. Reinmann<sup>1,2,3</sup>, Jonathan R. Thompson<sup>4</sup>, and David W. Allen<sup>5</sup>

<sup>1</sup>Department of Earth and Environment, Boston University, Boston, MA, USA, <sup>2</sup>Advanced Science Research Center at the Graduate Center, City University of New York, New York, NY, USA, <sup>3</sup>Department of Geography, Hunter College, New York, NY, USA, <sup>4</sup>Harvard Forest, Harvard University, Petersham, MA, USA, <sup>5</sup>National Institute of Standards and Technology, Gaithersburg, MD, USA

**Abstract** Forest fragmentation impacts carbon uptake and storage; however, the magnitude and direction of fragmentation impacts on soil respiration remain poorly characterized. We quantify soil respiration rates along edge-to-interior transects in two temperate broadleaf forests in the eastern United States that vary in climate, species composition, and soil type. We observe average soil respiration rates 15–26% higher at the forest edge compared to the interior, corresponding to large gradients in soil temperature. We find no significant difference in the sensitivity of soil respiration to temperature between the forest edge and interior. Fragmentation and resultant shifts in microenvironment alter forest productivity and soil respiration near forest edges. Ecosystem models do not currently represent edge dynamics, but given the prevalence of landscape fragmentation and its effect on carbon cycling along forest edges, this omission likely introduces an important source of uncertainty in our understanding of terrestrial carbon dynamics with a changing landscape and climate.

**Plain Language Summary** Forests continuously release carbon dioxide into the atmosphere via the decomposition of soil organic matter and biological activities of belowground organisms in a process known as soil respiration. When intact forests are broken up (either naturally or by humans), the remaining fragments experience distinct environmental conditions near the edges. Previous work shows that temperate forests can capture and store more atmospheric carbon per unit area near forest edges; however, it is unknown how the presence of forest edges impacts the rate of carbon loss via soil respiration. Here we measure soil respiration rates along forest edges, where the local soil temperatures tend to be warmer than the forest interior. We find that near the edges, soil respiration rates are higher than the forest interior. We attribute the increased soil respiration rate to warmer forest edge soil temperatures. Our results suggest that estimates of soil respiration in the temperate forest region may be underestimating biological emissions of carbon dioxide. Altogether, our research identifies an important phenomenon previously unaccounted for in our understanding of the carbon cycle with important implications for estimating forest carbon exchanges in fragmented landscapes, particularly as the climate continues to warm.

# 1. Introduction

Fragmented landscapes occupy a rapidly increasing portion of global land area, yet the impact of fragmentation on forest carbon fluxes has only recently begun to receive attention. More than 70% of the world's remaining forest area is located within 1 km of a forest edge (Haddad et al., 2015) and in the northeastern United States, nearly 25% of forest area is within 30 m of an edge (Smith et al., 2018). Forest fragmentation produces microenvironmental gradients near forest edges, commonly referred to as edge effects, that include increased exposure to solar radiation (Davies-Colley et al., 2000; Matlack, 1993), warmer temperatures (Chen et al., 1993), and drier conditions (Gehlhausen et al., 2000). Within the temperate forest region, the most fragmented forest biome in the world (Haddad et al., 2015), edge effects enhance the carbon accumulation rate and stocks in aboveground biomass with aboveground forest productivity and biomass reported to increase 89% and 64%, respectively, within 20 m of forest edges (Reinmann & Hutyra, 2017). However, the impact of edge effects on soil respiration rates is poorly understood.

Soils store the majority of the world's terrestrial organic carbon, and soil carbon fluxes act as a major control on atmospheric  $CO_2$  concentrations (Jobbágy & Jackson, 2000). Soil respiration represents the aggregate

 $CO_2$  flux from belowground autotrophic (plant roots) and heterotrophic (soil microorganisms and macroorganisms) processes and constitutes the second largest land-atmosphere carbon exchange behind gross primary productivity (Raich & Schlesinger, 1992). Soil respiration rates are primarily controlled by soil temperature (Lloyd & Taylor, 1994) but are also regulated by soil moisture content (Davidson et al., 1998), ecosystem productivity (Högberg et al., 2001), and substrate quality/quantity (Wan & Luo, 2003). Forest edge effects likely impact belowground carbon dynamics via gradients in local soil conditions that include warmer temperatures (Chen et al., 1993), drier soils (Pohlman et al., 2009), and increases in carbon and nitrogen availability (Remy et al., 2016).

This study quantifies the impact of edge effects on soil respiration rates in temperate forest fragments by testing the hypothesis that increases in soil temperature with proximity to forest edges stimulate soil respiration rates. We measure soil respiration fluxes coincident with soil temperature and moisture over multiple years along edge-to-interior transects in Massachusetts and Maryland, USA. We compare the sensitivity of soil respiration to soil temperature between the forest edge and interior and model soil respiration rates over the course of a year.

### 2. Materials and Methods

#### 2.1. Site Descriptions

#### 2.1.1. Harvard Forest

The Harvard Forest (HF) is a temperate, mixed deciduous forest located in Petersham, Massachusetts. The forest is largely made up of 85 to 120 years old, second-growth stands that followed widespread harvesting and agricultural clearing in the 1800s (Foster & Aber, 2004). In May 2016, six  $600\text{-m}^2$  plots were installed along forest edges at the HF, measuring 20 m along the forest edge and extending 30 m into the forest perpendicular to the forest edge (Figures 1a and 1c). The plots were established in well-drained, *Quercus sp./ Acer sp.*-dominated stands. Soils were developed on glacial till deposits and are predominantly mapped as Woodbridge series coarse loam soils with little clay content (Natural Resources Conservation Service, 2018). Mean summer (June, July, and August) air temperatures are 19.5 °C and mean winter (December, January, and February) air temperatures are -3.2 °C (Boose, 2018).

The plots were installed at multiple edge aspects and adjacent land cover types (three meadows, two pastures, and one road). Within each plot, a pair of polyvinyl chloride soil respiration collars 20 cm in diameter  $\times$  7 cm tall and located 10 m apart was inserted approximately 2 cm into the soil at four distances from the edge (0, 10, 20, and 30 m; Figure S1 in the supporting information). Each plot had n = 8 collars for a total of n = 48 collars. Following installation, collars were left in the soil for at least 2 weeks to equilibrate. Soil temperature and moisture at 10-cm depth were logged hourly using soil temperature (onset HOBO Pendant Datalogger) and soil moisture (onset Soil Moisture Smart Sensor) sensors located every 10 m along the center plot transect (i.e., 0, 10, 20, and 30 m from the forest edge).

During the growing seasons (May to October) of 2017 and 2018, measurements of soil respiration were made monthly in 2017 and weekly in 2018 using a LiCor LI-8100A soil respiration system with survey chamber (LiCor Biosciences). Respiration was measured between 8:00 and 16:00 local standard time; sampling order was varied to avoid sampling at the same time of day. Beginning in 2018, soil temperature and soil moisture probe measurements at 10-cm depth were recorded at the collar concurrent with the flux measurement (Hanna Instruments Thermistor Thermometer and Field Scout TDR 100, respectively). In addition to the logged soil temperature and moisture, 487 of the 977 observations of soil respiration at the HF included colocated point measurements of soil temperature and 362 included colocated point measurements of soil temperature.

#### 2.1.1.1. Additional HF Data

We complement our HF edge plot measurements of soil respiration with interior forest measurements made from 1995 to 2010 within the footprint of the Environmental Measurement Station's (EMS) eddy flux tower (42.537755°N, -72.171478°W; Davidson & Savage, 2017). The forest interior respiration data were collected at six sites along a northwest and southwest transect from the EMS tower. Sites ranged from poorly drained swamp to well-drained uplands but for the purposes of this comparison; we only use data collected at well-drained sites in mixed-deciduous forest. Measurements of soil respiration were taken weekly to biweekly with concurrent measurements of soil temperature at 10-cm depth and soil moisture at 15-cm depth. We



**Figure 1.** (a) Location of research plots at the Harvard Forest (HF) in Massachusetts, USA, and National Institute of Standards and Technology (NIST) in Maryland, USA. (b) Aerial view of 1-ha plot at NIST site. (c) Aerial view of one of six 600-m<sup>2</sup> HF plots. Imagery: ©2017 Google.

use hourly logged values of soil temperature and moisture at the EMS tower to compare our observations of hourly soil temperature at the edge and to predict hourly respiration rates over the course of a year. **2.1.2. National Institute of Standards and Technology** 

In May 2017, a 100-m  $\times$  100-m plot was installed along the edge of a 26-ha forest fragment on the campus of the National Institute of Standards and Technology (NIST) in Gaithersburg, Maryland (Figure 1). This plot was established as a Forested Optical Reference for the Evaluation of Sensor Technology, with soil respiration providing a key ground validation for remote sensing technologies. The plot is in a moderately well-drained, *Liriodendron tulipifera/Quercus sp.*-dominated stand with a 3-m-wide stream flowing through the plot approximately 25 m from the forest edge. Soils are mapped as a Glenelg series silt loam with a high clay content and low levels of soil organic matter (Natural Resources Conservation Service, 2018). Mean summer (June, July, and August) air temperatures are 23.5 °C and mean winter (December, January, and February) air temperatures are 1 °C (National Climatic Data Center, 2018).

The plot has a southeastern aspect and is adjacent to an unmanaged grassland (Figure 1b). Thirty-three pairs of 15.5-cm-diameter  $\times$  7-cm-tall polyvinyl chloride soil collars were placed in duplicate at five distances (0, 12.5, 25, 50, and 75 m; Figure S1) from the forest edge for a total sample size of n = 66 collars. Soil



**Figure 2.** Mean growing season (May to October) soil respiration rates ( $\mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) as a function of distance from the forest edge (m) at the National Institute of Standards and Technology (NIST) and Harvard Forest (HF). Error bars are 95% confidence intervals. See Figure S3 for histograms of observations in each site.

temperature and moisture at 10-cm depth were logged hourly using 22 soil temperature sensors (onset HOBO Pendant Datalogger) located at 11 locations throughout the plot and 20 soil moisture sensors (onset Soil Moisture Smart Sensor) located at 5 locations (Figure S1).

During the growing seasons (May to October) of 2017 and 2018, soil respiration measurements were made weekly to biweekly using a soil respiration backpack system with a  $CO_2$  gas analyzer (LiCor LI-840A) following the vented design described by Savage and Davidson (2001). Measurements were generally taken over the course of two consecutive days between 8:00 and 16:00 local standard time and the order collars were sampled in was varied. When possible, soil temperature and soil moisture probe measurements at 10-cm depth were recorded at the collar simultaneous with the flux measurement (Hanna Instruments Thermistor Thermometer and Field Scout TDR 100, respectively). Of the 1,713 observations of soil respiration at the NIST, 1,416 include colocated point measurements of soil temperature and 533 include colocated point measurements of soil temperature.

#### 2.2. Temperature Response and Modeling

We quantify the response of soil respiration to soil temperature using a nonlinear regression model in the form:

$$Rs = \beta_0 * \exp^{(\beta_1 * Ts)},\tag{1}$$

where *R*s is the soil respiration rate ( $\mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) and *T*s is the soil temperature at 10-cm depth (°C). The *Q*<sub>10</sub>, or the factor by which soil respiration increases with a 10 °C change in soil temperature (Davidson & Janssens, 2006), is calculated as



**Figure 3.** (top) Relationship between soil respiration rates ( $\mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) and soil temperature (°C) at 10-cm depth at various distances from the edge at the Harvard Forest (a) and National Institute of Standards and Technology (b). Model fits are nonlinear regression models and shaded error regions represent 95% confidence intervals of the fitted parameters  $\beta_0$  and  $\beta_1$  in equation (1). (bottom) Frequency distribution of hourly logged soil temperature (°C) during the growing seasons (May to October) of 2017/2018 at the Harvard Forest (c) and National Institute of Standards and Technology (d). EMS = Environmental Measurement Station.

$$Q_{10} = \exp^{(10 * \beta_1)}$$
 (2)

We use a generalized additive model and hourly measurements of local soil conditions to predict soil respiration rates in each site at various distances from the edge between August 2017 and July 2018. The structure of the model is

$$\ln (Rs) = \beta + s (Ts) + s (VWC) + \varepsilon, \qquad (3)$$

where  $\beta$  is the intercept of the model, *s* (*Ts*) is a smooth spline function of soil temperature (°C), *s* (VWC) is a smooth spline function of volumetric water content (%), and  $\varepsilon$  is the Gaussian distributed error of the model (see Table S1 and Figure S2 for detailed model diagnostics and parameter outputs). Previous work shows that annual soil respiration fluxes can be estimated based on year-round measurements of soil temperature/moisture (Davidson et al., 1998; Giasson et al., 2013; Rey et al., 2002).

We estimate hourly soil respiration rates at the HF using the average continuously logged soil temperature and moisture at 0 and 30 m from the forest edge across the six edge plots and continuously logged soil temperature and moisture at the HF EMS tower (>200 m from the nearest edge). NIST hourly estimates use continuously logged soil temperature and moisture data collected at 0 and 75 m from the southeastern facing edge of the plot. Hourly flux estimates are summed to compute annual carbon fluxes in units of megagrams of carbon per hectare per year. Unless otherwise noted, we refer to the forest edge as the start of the forest





**Figure 4.** (top) Modeled hourly soil respiration rates ( $\mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) at various distances from the forest edge at the Harvard Forest (a) and National Institute of Standards and Technology (b). (bottom) Modeled annual soil carbon flux (Mg·ha<sup>-1</sup>·year<sup>-1</sup>) at various distances from the forest edge at the Harvard Forest (c) and National Institute of Standards and Technology (d). Error bars are 95% confidence intervals.

edge transects (i.e., 0 m from the forest edge) and the forest interior as the furthest point along the transects (30 m from the forest edge at the HF and 75 m from the forest edge at the NIST site). All reported errors are 95% confidence intervals and all statistical analyses and modeling were conducted using R Statistical Software 3.5 (R Core Team, 2018).

#### 3. Results

#### 3.1. Forest Edge Gradients

We observe strong positive gradients in soil respiration with proximity to forest edges. Mean growing season respiration rates were 26% higher at the forest edge ( $8.2 \pm 0.5 \mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) than 30 m into the forest interior ( $6.5 \pm 0.3 \mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>; p < 0.01) at the HF. At the NIST, overall soil respiration rates were nearly 70% lower than at the HF, but mean growing season respiration rates showed a similar trend with proximity to the edge, with respiration rates 15% higher at the forest edge ( $2.3 \pm 0.1 \mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>) than 75 m into the forest interior ( $2.0 \pm 0.1 \mu$ mol CO<sub>2</sub>·m<sup>-2</sup>·s<sup>-1</sup>; p < 0.01; Figure 2). While we observe elevated soil respiration rates only at the forest edge at the HF, we observe significant soil respiration enhancements up to at least 12.5 m into the forest at the NIST (Figure 2).

#### 3.2. Temperature Response

There was no significant difference in the  $Q_{10}$  response between the forest edge and interior at either the HF (p = 0.59) or NIST (p = 0.66). At the HF, we observed a  $Q_{10}$  of  $3.2 \pm 0.4$  within 30 m of an edge and  $3.3 \pm 0.4$  at the EMS tower (Figure 3a). At the NIST, we observed a lower mean  $Q_{10}$  value than HF, that is, soil

respiration rates were less sensitive to soil temperatures, with values of  $1.9 \pm 0.2$  within 12.5 m of an edge and  $1.8 \pm 0.2$  beyond 12.5 m from the edge (Figure 3b).

At both sites, we observed a greater frequency of high soil temperatures during the growing season at the forest edge than in the forest interior (Figures 3c and 3d). At the HF, mean growing season soil temperatures at the forest edge  $(16.4 \pm 0.1 \text{ °C})$  were 9% higher than 30 m into the forest interior  $(15.0 \pm 0.1 \text{ °C}; p < 0.01)$  and 12% higher than at the EMS tower  $(14.6 \pm 0.1 \text{ °C}; p < 0.01;$  Figure 3c). At the NIST, mean growing season soil temperatures at the forest edge  $(19.6 \pm 0.1 \text{ °C}; p < 0.01;$  Figure 3c). At the NIST, mean growing season soil temperatures at the forest edge  $(19.6 \pm 0.1 \text{ °C})$  were 5% higher than 75 m into the forest interior  $(18.7 \pm 0.1 \text{ °C}; p < 0.01;$  Figure 3d). All of our observations of soil respiration were made during the day, but nighttime forest interior soil temperatures were often higher than nighttime forest edge soil temperatures (Figure S4).

#### 3.3. Annual Fluxes

We find considerable increases in annual soil carbon fluxes at the forest edge. Using a generalized additive model relating soil respiration rates to soil temperature and moisture (model results in Table S1 and Figure S2), we estimate the annual soil carbon flux at the forest edge at the HF ( $13.3 \pm 0.8$  Mg C·ha<sup>-1</sup>·year<sup>-1</sup>) to be 32% higher than the annual soil carbon flux 30 m from the edge ( $10.1 \pm 0.6$  Mg C·ha<sup>-1</sup>·year<sup>-1</sup>; p < 0.01) and 41% higher than the annual soil carbon flux at the EMS tower ( $9.4 \pm 0.5$  Mg C·ha<sup>-1</sup>·year<sup>-1</sup>; p < 0.01; Figures 4a and 4c). Our modeled estimates agree well with annual ecosystem respiration estimates from the HF EMS eddy flux tower if we assume soil respiration accounts for approximately 70% of ecosystem respiration (Bolstad et al., 2004; Yuste et al., 2005) as the mean annual ecosystem respiration rate over the 10 most recent years of available data (2006 to 2015) is 13.7 Mg C·ha<sup>-1</sup>·year<sup>-1</sup> (Munger & Wofsy, 2018). At the NIST, we estimate the annual soil carbon flux at the forest edge ( $4.3 \pm 0.4$  Mg C·ha<sup>-1</sup>·year<sup>-1</sup>; p < 0.01; Figures 4b and 4d).

### 4. Discussion and Conclusion

## 4.1. Temperate Forest Edge Carbon Balance

Forests are a globally important carbon sink, offsetting nearly one third of anthropogenic carbon emissions (Le Quéré et al., 2018). The strength of the forest carbon sink varies spatially and temporally and depends on climatic trends, disturbances, and the resultant balance between forest growth, mortality, and respiration dynamics. In response to a changing climate, soil warming experiments provide strong empirical support for the hypothesis that increases in global temperatures will stimulate the rate of soil carbon losses, driving a positive carbon-climate feedback that could accelerate rates of global warming (Crowther et al., 2016). Furthermore, fragmentation represents a key landscape disturbance and the observations of soil respiration and temperature gradients with proximity to forest edges described in this study add to a growing body of literature pointing to the high capacity of forest fragmentation to alter forest carbon cycling (Bowering et al., 2006; Briber et al., 2015; Brinck et al., 2017; Reinmann & Hutyra 2017).

Positive soil respiration gradients near temperate forest edges offset a fraction of observed enhancements in aboveground production and biomass. However, while not a direct comparison, our estimates of annual forest edge soil respiration increases of 32% and 13% at the HF and NIST, respectively, are likely not large enough to offset the 89% aboveground net carbon uptake enhancement observed in nearby temperate forest fragments (Reinmann & Hutyra, 2017). Soil respiration is the largest of several contributors to ecosystem respiration (Bolstad et al., 2004), and it is likely that increases in biomass and forest growth coupled with warmer air temperatures near temperate forest edges will also positively correlate with increases in above-ground autotrophic respiration (Reinmann & Templer, 2016) offsetting even more of the edge growth enhancement. Nonetheless, due to enhancements in both respiration and growth, carbon cycling rates currently appear to accelerate near temperate forest edges.

The forest edge is warmer than the forest interior on average, but we find the forest edge tends to lose heat faster than the forest interior at night. Warmer nighttime forest interior soils likely offset a portion of the daytime edge respiration enhancement when considering the net forest edge soil respiration enhancement over the course of the year. We observe warmer soil temperatures in the forest interior during 20% and 55% of growing season nights at the HF and NIST, respectively.

Forest fragmentation increases the proportion of forest edge across the landscape, heightening the vulnerability of the remaining forest to climate change and extreme conditions. The forest edge is warmer and drier than the forest interior (Davies-Colley et al., 2000) and growth at the forest edge has been reported to decline three times faster in response to heat stress than the forest interior (Reinmann & Hutyra, 2017). As such, future climate projections of warmer temperatures and increases in the frequency and severity of droughts (IPCC (Intergovernmental Panel on Climate Change), 2014) are forecasted to diminish the forest edge growth enhancement (Reinmann & Hutyra, 2017). In contrast, we observe strong exponential increases in soil respiration with warmer temperatures (Figure 3) and as increasing trends are already being observed in global soil carbon fluxes (Bond-Lamberty & Thompson, 2010), our results suggest that respiration rates will likely increase at an accelerated rate near the forest edge than the forest interior. Although temperature is the primary control on soil respiration, drought conditions can decrease respiration rates (Davidson et al., 1998) and could potentially mitigate a portion of temperature associated increases in soil respiration.

#### 4.2. Implications for Carbon Accounting

Much of our current understanding of forest carbon fluxes comes from the study of intact forest systems, particularly relying on eddy covariance measurements. The fundamental aerodynamic assumptions of eddy covariance methodologies require homogeneity within the fetch of the eddy flux tower (Lee et al., 2006), introducing a strong selection bias against fragmented landscapes and the distinct carbon dynamics found within them. Despite pervasive global fragmentation and substantial impacts of fragmentation on forest carbon dynamics, carbon cycle models and accounting initiatives currently lack the necessary empirical data to parameterize and model forest edge dynamics at large scales. Edge dynamics are generally omitted from estimates of forest carbon pools and fluxes. Our estimates of increased soil carbon fluxes of 3.2-3.9 Mg C·ha  $^{-1}$ ·year<sup>-1</sup> at the HF edges are larger than the average net ecosystem production within the footprint of the EMS eddy flux tower (2.9 Mg C·ha<sup>-1</sup>·year<sup>-1</sup>; Munger & Wofsy, 2018). Soil respiration enhancements of such magnitude could be enough to shift the balance of some forests from a carbon sink to a carbon source near edges.

Partitioning the forest edge respiration response between autotrophic and heterotrophic processes is unclear. At the HF, autotrophic respiration only accounts for about 20% of total soil respiration (Melillo et al., 2002) and to our knowledge, the only study quantifying root biomass near forest edges found no significant trends in root biomass with proximity to the edge (Reinmann & Hutyra, 2017). Constant root biomass coupled with observed positive relationships between photosynthesis and soil respiration (Tang et al., 2005) suggest that enhanced forest productivity could increase the contributions of autotrophic processes near the forest edge. Further, parsing the respective roles of roots versus soil microbes in the forest edge soil respiration enhancement will have important implications for our understanding of the interactions of edge effects with climate change and the extent to which forest edge respiration is coupled with productivity.

The drivers and intensity of forest fragmentation could also influence the forest edge respiration response. While we observe large enhancements of soil respiration in rural (HF) and suburban (NIST) forest fragments, the response in urban forest fragments may differ due to enhanced atmospheric nitrogen inputs (Decina et al., 2017; Rao et al., 2013). Atmospheric nitrogen deposition can be more than 40% higher at the forest edge than the forest interior (Weathers et al., 2001). Increased soil nitrogen availability has been shown to decrease both heterotrophic and autotrophic soil respiration rates (Janssens et al., 2010; Sun et al., 2014). Consequently, the respiration dynamics along forest edges and net forest carbon balance in industrial regions may differ from those of rural forests.

Forest edge effects on carbon dynamics also vary by biome (Smith et al., 2018). To our knowledge, this is the first study quantifying the impact of edge effects on soil respiration rates, and there exists considerable uncertainty and probable spatial variability in respiration responses across forest types (Giasson et al., 2013). In our study alone, we observe large differences in  $Q_{10}$  factors between the cooler HF site and warmer NIST site, corresponding to a larger respiration enhancement at the HF. Previous studies show that  $Q_{10}$  factors are generally higher in cooler regions (Chen & Tian, 2005) indicating that the magnitude of the edge respiration response may vary with latitude. Similarly, the enhancements in forest edge productivity and biomass observed in the temperate region contradict trends in forest edge carbon storage in boreal (Jönsson et al., 2007) and tropical forests (Laurance et al., 1997).

Our findings indicate that forest fragmentation and edge effects introduce large changes in the soil microenvironment that produce significant changes in overall soil respiration rates. This study reinforces the concept of a distinct forest edge environment as forest fragmentation simultaneously stimulates forest productivity and respiration in temperate forests, with important implications for carbon balance accounting. While typically not considered, the indirect effects of land cover change and deforestation on carbon cycling (e.g., edge effects) are likely introducing an unaccounted source of uncertainty in our estimates of the global carbon budget. Ubiquitous global fragmentation suggests that consideration of edge effects and their interactions with climate change is critical to accurately characterize the world's forests.

#### References

Bolstad, P. V., Davis, K. J., Martin, J., Cook, B. D., & Wang, W. (2004). Component and whole-system respiration fluxes in northern deciduous forests. *Tree Physiology*, 24(5), 493–504. https://doi.org/10.1093/treephys/24.5.493

Bond-Lamberty, B., & Thompson, A. (2010). Temperature-associated increases in the global soil respiration record. *Nature*, 464, 579–582. https://doi.org/10.1038/nature08930

Boose, E. (2018). Fisher Meteorological Station at Harvard Forest since 2001. Harvard Forest Data Archive: HF001.

- Bowering, M., Lemay, V., & Marshall, P. (2006). Effects of forest roads on the growth of adjacent lodgepole pine trees. Canadian Journal of Forest Research, 36(4), 919–929. https://doi.org/10.1139/x05-300
- Briber, B. M., Hutyra, L. R., Reinmann, A. B., Raciti, S. M., Dearborn, V. K., Holden, C. E., & Dunn, A. L. (2015). Tree productivity enhanced with conversion from forest to urban land covers. *PLoS ONE*, *10*(8), e0136237. https://doi.org/10.1371/journal.pone.0136237 Brinck, K., Fischer, R., Groeneveld, J., Lehmann, S., Paula, M. D., Pütz, S., et al. (2017). High resolution analysis of tropical forest frag-
- mentation and its impact on the global carbon cycle. *Nature Communications*, 8, 14855. https://doi.org/10.1038/ncomms14855 Chen, H., & Tian, H. (2005). Does a general temperature-dependent  $Q_{10}$  model of soil respiration exist at biome and global scale? *Journal of*
- Integrative Plant Biology, 47(11), 1288–1302. https://doi.org/10.1111/j.1744-7909.2005.00211.x Chen, J., Franklin, J. F., & Spies, T. A. (1993). Contrasting microclimates among clearcut, edge, and interior of old-growth Douglas-fir forest. Agricultural and Forest Meteorology, 63(3–4), 219–237. https://doi.org/10.1016/0168-1923(93)90061-l
- Crowther, T. W., Todd-Brown, K. E. O., Rowe, C. W., Wieder, W. R., Carey, J. C., Machmuller, M. B., et al. (2016). Quantifying global soil carbon losses in response to warming. *Nature*, 540(7631), 104–108. https://doi.org/10.1038/nature20150
- Davidson, E., & Savage, K. (2017). Soil respiration, temperature and moisture at Harvard Forest EMS Tower since 1995. Harvard Forest Data Archive: HF006. https://doi.org/10.6073/pasta/33ba3432103297fe0644de6e0898f91f
- Davidson, E. A., Belk, E., & Boone, R. D. (1998). Soil water content and temperature as independent or confounded factors controlling soil respiration in a temperate mixed hardwood forest. *Global Change Biology*, 4(2), 217–227. https://doi.org/10.1046/j.1365-2486.1998.00128.x
- Davidson, E. A., & Janssens, I. A. (2006). Temperature sensitivity of soil carbon decomposition and feedbacks to climate change. Nature, 440(7081), 165–173. https://doi.org/10.1038/nature04514
- Davies-Colley, R. J., Payne, G. W., & van Elswijk, M. (2000). Microclimatic gradients across a forest edge. New Zealand Journal of Ecology, 24(2), 111–121.
- Decina, S. M., Templer, P. H., Hutyra, L. R., Gately, C. K., & Rao, P. (2017). Variability, drivers, and effects of atmospheric nitrogen inputs across an urban area: Emerging patterns among human activities, the atmosphere, and soils. *Science of the Total Environment*, 609, 1524–1534. https://doi.org/10.1016/j.scitotenv.2017.07.166
- Foster, D. R., & Aber, J. D. (2004). Forests in time: The environmental consequences of 1,000 years of change in New England. New Haven, CT: Yale University Press.
- Gehlhausen, S. M., Schwartz, M. W., & Augspurger, C. K. (2000). Vegetation and microclimatic edge effects in two mixed-mesophytic forest fragments. *Plant Ecology*, 147(1), 21–35. https://doi.org/10.1023/A:1009846507652
- Giasson, M. –. A., Ellison, A. M., Bowden, R. D., Crill, P. M., Davidson, E. A., Drake, J. E., et al. (2013). Soil respiration in a northeastern US temperate forest: A 22-year synthesis. *Ecosphere*, 4(11), art140–art128. https://doi.org/10.1890/ES13.00183.1

Haddad, N. M., Brudvig, L. A., Clobert, J., Davies, K. F., Gonzalez, A., Holt, R. D., et al. (2015). Habitat fragmentation and its lasting impact on Earth's ecosystems. *Science Advances*, 1(2). https://doi.org/10.1126/sciadv.1500052

- Högberg, P., Nordgren, A., Buchmann, N., Taylor, A. F. S., Ekblad, A., Högberg, A. N., et al. (2001). Large-scale forest girdling shows that current photosynthesis drives soil respiration. *Nature*, 411, 78–792.
- IPCC (Intergovernmental Panel on Climate Change) (2014). Climate change 2014: Synthesis report. Contribution of Working Groups I, II, and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva, Switzerand: IPCC.
- Janssens, I. A., Dieleman, W., Luyssaert, S., Subke, J.-A., Reichstein, M., Ceulemans, R., et al. (2010). Reduction of forest soil respiration in response to nitrogen deposition. *Nature Geoscience*, *3*, 315–322. https://doi.org/10.1038/ngeo844
- Jobbágy, E. G., & Jackson, R. B. (2000). The vertical distribution of soil organic carbon and its relation to climate and vegetation. *Ecological Applications*, 10(2), 423–436. https://doi.org/10.1890/1051-0761(2000)010[0423:TVDOSO]2.0.CO;2
- Jönsson, M. T., Fraver, S., Jonsson, B. G., Dynesius, M., Rydgård, M., & Esseen, P. (2007). Eighteen years of tree mortality and structural change in an experimentally fragmented Norway spruce forest. Forest Ecology and Management, 242(2–3), 306–313. https://doi.org/ 10.1016/j.foreco.2007.01.048
- Laurance, W. F., Laurance, S. G., Ferreira, L. V., Rankin-de Merona, J. M., Gascon, C., & Lovejoy, T. E. (1997). Biomass collapse in amazonian forest fragments. Science, 278(5340), 1117–1118. https://doi.org/10.1126/science.2785340.1117
- Le Quéré, C., Andrew, R. M., Friedlingstein, P., Sitch, S., Pongratz, J., Manning, A. C., et al. (2018). Global carbon budget 2017. Earth System Science Data, 10, 405–448. https://doi.org/10.5194/essd-10-405-2018
- Lee, X., Massman, W., & Law, B. (2006). Handbook of micrometeorology: A guide for surface flux measurement and analysis. Berlin, Germany: Springer Science & Business Media.
- Lloyd, J., & Taylor, J. A. (1994). On the temperature dependence of soil respiration. Functional Ecology, 8(3), 315. https://doi.org/10.2307/2389824
- Matlack, G. R. (1993). Microenvironment variation within and among forest edge sites in the eastern United States. *Biological Conservation*, 66(3), 185–194. https://doi.org/10.1016/0006-3207(93)90004-k

#### Acknowledgments Harvard Forest EMS soil respiration

data were provided by Eric Davidson and Kathleen Savage. Harvard Forest EMS tower data were provided by J. William Munger and Steven Wofsy. Harvard Forest flux measurements are supported by the AmeriFlux Management Project with funding by the U.S. Department of Energy's Office of Science under Contract DE-AC02-05CH11231 and additionally as a component of the Harvard Forest LTER site supported by the National Science Foundation (DEB-1237491). We thank Lauren Cabrera, Sarah Garvey, Jonathan Gewirtzman, Wiley Hundertmark, Erin Pierce, and Joy Winbourne for their help in the field and Ian Sue Wing for helpful insights in data analysis. Financial support for this research was provided by the United States Department of Agriculture National Institute of Food and Agriculture Award 2017-67003-26615 and National Institute of Standards and Technology Award 70NANB17H030. Data reported in this paper are available at Dataverse (doi: http://10.7910/DVN/ ZLRKK2). Certain commercial equipment, instruments, or materials (suppliers or software) are identified in this paper to foster understanding. Such identification does not imply recommendation or endorsement by the National Institute of Standards and Technology nor does it imply that the materials or equipment identified are necessarily the best available for the purpose.

- Melillo, J., Steudler, P., Aber, J., Newkirk, K., Lux, H., Bowles, F., et al. (2002). Soil warming and carbon-cycle feedbacks to the climate system. Science, 298(5601), 2173–2176. https://doi.org/10.1126/science.1074153
- Munger, W., Wofsy, S. (2018). Canopy-atmosphere exchange of carbon, water and energy at Harvard Forest EMS Tower since 1991. Harvard Forest Data Archive: HF004.
- National Climatic Data Center (2018). Climate data online: Data tools, www.ncdc.noaa.gov/cdo-web/datatools/normals (Last accessed December 2, 2018).
- Natural Resources Conservation Service (2018). Web soil survey, www.websoilsurvey.sc.egov.usda.gov/App/HomePage.htm (Last accessed November 29, 2018).
- Pohlman, C. L., Turton, S. M., & Goosem, M. (2009). Temporal variation in microclimatic edge effects near powerlines, highways and streams in Australian tropical rainforest. Agricultural and Forest Meteorology, 149(1), 84–95. https://doi.org/10.1016/j. agrformet.2008.07.003
- R Core Team (2018). R: A language and environment for statistical computing. R foundation for Statistical Computing, Vienna, Austria. Retrieved from http://www.R-project.org
- Raich, J. W., & Schlesinger, W. H. (1992). The global carbon dioxide flux in soil respiration and its relationship to vegetation and climate. *Tellus B: Chemical and Physical Meteorology*, 44(2), 81–99. https://doi.org/10.3402/tellusb.v44i2.15428

Rao, P., Hutyra, L. R., Raciti, S. M., & Templer, P. H. (2013). Atmospheric nitrogen inputs and losses along an urbanization gradient from Boston to Harvard Forest, MA. *Biogeochemistry*, 121(1), 229–245. https://doi.org/10.1007/s10533-013-9861-1

- Reinmann, A. B., & Hutyra, L. R. (2017). Edge effects enhance carbon uptake and its vulnerability to climate change in temperate broadleaf forests. Proceedings of the National Academy of Sciences, 114(1), 107–112. https://doi.org/10.1073/pnas.1612369114
- Reinmann, A. B., & Templer, P. H. (2016). Reduced winter snowpack and greater soil frost reduce live root biomass and stimulate radial growth and stem respiration of red maple (*Acer rubrum*) trees in a mixed-hardwood forest. *Ecosystems*, 19(1), 129–141. https://doi.org/ 10.1007/s10021-015-9923-4

Remy, E., Wuyts, K., Boeckx, P., Ginzburg, S., Gundersen, P., Demey, A., et al. (2016). Strong gradients in nitrogen and carbon stocks at temperate forest edges. *Forest Ecology and Management*, 376, 45–58. https://doi.org/10.1016/j.foreco.2016.05.040

Rey, A., Pegorano, E., Tedeschi, V., De Parri, I., Jarvis, P. G., & Velentinit, R. (2002). Annual variation in soil respiration and its components in a coppice oak forest in Central Italy. *Global Change Biology*, 8(9), 851–866. https://doi.org/10.1046/j.1365-2486.2002.00521.x

Savage, K. B., & Davidson, E. A. (2001). Interannual variation of soil respiration in two New England forests. *Global Biogeochemical Cycles*, 15(2), 337–350. https://doi.org/10.1029/1999GB001248

Smith, I. A., Hutyra, L. R., Reinmann, A. B., Marrs, J. K., & Thompson, J. R. (2018). Piecing together the fragments: Elucidating edge effects on forest carbon dynamics. Frontiers in Ecology and the Environment, 16(4), 213–221. https://doi.org/10.1002/fee.1793

Sun, Z., Liu, L., Ma, Y., Yin, G., Zhao, C., Zhang, Y., & Piao, S. (2014). The effect of nitrogen addition on soil respiration from a nitrogenlimited forest soil. Agricultural and Forest Meteorology, 197, 103–110. https://doi.org/10.1016/j.agrformet.2014.06.010

Tang, J., Baldocchi, D. D., & Xu, L. (2005). Tree photosynthesis modulates soil respiration on a diurnal time scale. *Global Change Biology*, *11*, 1298–1304. https://doi.org/10.1111/j.1365-2486.2005.00978.x

- Wan, S., & Luo, Y. (2003). Substrate regulation of soil respiration in a tallgrass prairie: Results of a clipping and shading experiment. Global Biogeochemical Cycles, 17(2), 1054. https://doi.org/10.1029/2002GB001971
- Weathers, K. C., Cadenasso, M. L., & Pickett, S. T. (2001). Forest edges as nutrient and pollutant concentrators: Potential synergisms between fragmentation, forest canopies, and the atmosphere. *Conservation Biology*, *15*(6), 1506–1514. https://doi.org/10.1046/j.1523-1739.2001.01090.x
- Yuste, J. C., Nagy, M., Janssens, I. A., Carrara, A., & Ceulemans, R. (2005). Soil respiration in a mixed temperate forest and its contribution to total ecosystem respiration. *Tree Physiology*, 25(5), 609–619. https://doi.org/10.1093/treephys/25.5.609